Author’s response to reviews

Title: The risk of cesarean delivery after labor induction among women with prior pregnancy complications: A subgroup analysis of the AFFIRM study

Authors:

Leslie Skeith (leslie.skeith@gmail.com)
Grégoire Le Gal (glegal@ohri.ca)
Johanna de Vries (jip.devries@vumc.nl)
Saskia Middeldorp (s.middeldorp@amc.uva.nl)
Mariëtte Goddijn (m.goddijn@amc.uva.nl)
Risto Kaaja (riskaa@utu.fi)
Jean-Christophe Gris (jean.christophe.gris@chu-nimes.fr)
Ida Martinelli (martin@policlinico.mi.it)
Ekkehard Schleußner (ekkehard.schleussner@med.uni-jena.de)
David Petroff (david.petroff@zks.uni-leipzig.de)
Nicole Langlois (nlangois@ohri.ca)
Marc Rodger (mrodger@ohri.ca)

Version: 2 Date: 17 Nov 2019

Author’s response to reviews:

Dear Ahmed Abbas,

We have provided the corrections below:

1. Ethics approval and consent to participate

As the data is not publicly available, please state whether you received administrative permission to access and use the data analysed.

Response: We received administrative permission, which is now clarified in the manuscript.
2. Title page

We noticed that the corresponding author information provided on your title page is different to the one you indicated in the editorial submission. Please amend one so that they match.

Response: We confirm that the corresponding author information on the title page is correct.

3. Funding

In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: We added this.

4. Availability of data and material

Please could you confirm if it is possible to obtain an anonymised form of the data or who to contact for permissions to obtain the data.

Response: We have included this in the manuscript. Unfortunately, it is not possible to obtain an anonymised form of the data.

5. Acknowledgements

Please remove the funding information from the Acknowledgements and include it in the Funding section instead if this funding was directly involved in the study completion, if not include this information in your competing interests section. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.

Response: We have corrected this.

If you have any questions, please do not hesitate to contact myself.

Leslie Skeith, MD, FRCPC, MHPE

Division of Hematology & Hematological Malignancies
University of Calgary, Canada
Email: laskeith@ucalgary.ca
Phone: 403-944-5246
Fax: 403-944-1647